RESEARCH ARTICLE

# Surgical debulking of podoconiosis nodules and its impact on quality of life in Ethiopia

Wendemagegn Enbiale<sup>1,2</sup>\*, Kristien Verdonck<sup>3</sup>, Melesse Gebeyehu<sup>1</sup>, Johan van Griensven<sup>3</sup>, Henry J. C. de Vries<sup>2,4</sup>

1 Bahir Dar University, college of Medicine and Health sciences, Bahir Dar, Ethiopia, 2 Amsterdam UMC, University of Amsterdam, department of dermatology, Amsterdam Institute for Infection and Immunity (AII), location Academic Medical Centre, Amsterdam, The Netherlands, 3 Institute of Tropical Medicine, Antwerp, Belgium, 4 STI outpatient clinic, Department of Infectious Diseases, Public Health Service Amsterdam, The Netherlands

\* wendemagegnenbiale@gmail.com

# Abstract

# Background

In Ethiopia, severe lymphedema and acute dermato-lymphangio-adenitis (ADLA) of the legs as a consequence of podoconiosis affects approximately 1.5 million people. In some this condition may lead to woody-hard fibrotic nodules, which are resistant to conventional treatment. We present a series of patients who underwent surgical nodulectomy in a resource-limited setting and their outcome.

# Methods

In two teaching hospitals, we offered surgical nodulectomies under local anaesthesia to patients with persisting significant fibrotic nodules due to podoconiosis. Excisions after nodulectomy were left to heal by secondary intention with compression bandaging. As outcome, we recorded time to re-epithelialization after surgery, change in number of ADLA episodes, change in quality of life measured with the Dermatology Quality of Live Index (DQLI) questionnaire, and recurrence rate one year after surgery.

#### Results

37nodulectomy operations were performed on 21 patients. All wounds re-reepithelialised within 21 days (range 17–42). 4 patients developed clinically relevant wound infections. The DLQI values were significantly better six months after surgery than before surgery (P<0.0001). Also the number of ADLA episodes per three months was significantly lower six months after surgery than before surgery (P<0.0001).

# Conclusion

Nodulectomy in podoconiosis patients leads to a significant improvement in the quality of life with no serious complications, and we recommend this to be a standard procedure in resource-poor settings.



# G OPEN ACCESS

**Citation:** Enbiale W, Verdonck K, Gebeyehu M, van Griensven J, de Vries HJC (2021) Surgical debulking of podoconiosis nodules and its impact on quality of life in Ethiopia. PLoS Negl Trop Dis 15(1): e0009053. https://doi.org/10.1371/journal. pntd.0009053

Editor: José Antonio Suárez, Instituto Conmemorativo Gorgas de Estudios de la Salud, PANAMA

Received: October 17, 2020

Accepted: December 8, 2020

Published: January 22, 2021

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0009053

**Copyright:** © 2021 Enbiale et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript.

**Funding:** The author(s) received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

# Author summary

Podoconiosis is a disease that causes swelling of the lower legs and tissue lumps (nodules) of the feet. The condition is physically disabling with significant psycho-social impact. It is caused by the destruction of the lymphatic system, that is critical in the transportation of body fluids. The disease is said to be caused by long term barefooted exposure to red-clay soil. Podoconiosis affects more than four million subsistent farmers in the tropics. Ethiopia is one of the endemic countries with more than 1.5 million affected people.

Early-stage podoconiosis morbidity management consists of foot hygiene, regular inspection and care for traumatic abrasions, use of moisturizer, daily exercise, compression with elastic bandages, and last but not least shoe wearing. In the late disease stage, irreversible swelling and woody-hard fibrous nodules arise. As a result, patients can no longer wear shoes and suffer from disabling bacterial infection. Both complications aggravate lymphatic destruction. Moreover, late-stage podoconiosis is unresponsive to the aforementioned therapy options. Surgical intervention removing the nodules (nodulectomy) can potentially alleviate late-stage disease.

Here, we describe the clinical outcome of nodulectomy and its effect on the quality of life in 21 late stage podoconiosis patients, in a before to 12 months after the procedure comparison. We demonstrate that nodulectomy is safe, allows patients to wear shoes again, reduces the number of bacterial infections, and significantly improves their quality of life. Nodulectomy is a safe and effective procedure for podoconiosis patients. We recommend nodulectomy as the preferred therapy option for late-stage podoconiosis complicated by nodules.

# Background

Podoconiosis, or non-filarial elephantiasis is one of the causes of lymphedema. The disease affects barefoot subsistence farmers exposed to soil originating from volcanic rock in wet rural highland areas, in a population with genetic susceptibility [1–5]. Globally, podoconiosis affects an estimated 4 million people in about 17 countries, mostly in areas of tropical Africa[6].

In Ethiopia, about one third of the districts in the country are endemic for podoconiosis, with more than 35 million people at risk[7]. Within these endemic areas, population-based surveys have produced prevalence estimates of 5–10%. Overall, podoconiosis is thought to affect about 1.5 million people in the country [7]. The disease is physically disabling and has severe economic consequences: productivity losses per patient amount to 45% and sum up to huge economic losses at country level [8,9]. Stigmatization of people with podoconiosis is marked: some patients are excluded from social gatherings, school, churches and mosques, and are barred from marriage with unaffected individuals [10].

On clinical examination, these patients often have a dry, hyperkeratotic skin with deep cracks in the heels and between the toes. Many patients go on to develop gross oedema and fibrotic nodular and tumorous masses [11,12]. Untreated patients suffer episodes of acute dermato-lymphangio-adenitis(ADLA), with an associated risk of morbidity and mortality[13]. Further more, clinical observation showed that recurrent ADLA results in exacerbation of the lymphedema. The severity of the disease can be classified using a locally modified, simple, five-stage method based on the level of oedema, presence of nodules/tumours, and fixation of the joint [14].

Edema may occur in two forms: pitting or non-pitting (woody hard and fibrotic). The nonpitting form is especially troublesome for the patient because it is associated with multiple, fibrotic skin nodules that cause pain on shoe wearing, and that are easily traumatized leading to ulceration. Moreover, in the non-pitting form of edema, the skin folds are prone to maceration and fungal and bacterial infection [12]. This results in a very low quality of life for patients with nodules [15].

Conventional therapy consists of foot hygiene, regular inspection and care for traumatic abrasions, use of moisturizer, daily exercise, leg elevation, shoe wearing and compression with short-stretched elastic bandages. Although rigorous evidence is lacking, such conventional preventive therapy seems to improve the lymphedema and stop the recurrent ADLA for most patients with mild forms of podoconiosis (stage I and II lymphedema)[13,16]. By contrast, patients with non-pitting oedema and woody-hard fibrous nodules on foot and leg tend to have a poor response to conventional therapy. The oedema and nodules do not respond to compression therapy and the patients have frequently recurring episodes of ADLA. Removal of the nodules in these patients could allow them to make better use of the conventional therapy and decrease the episodes of ADLA[17].

Previously, we reported on11 patients with podoconiosis nodules who underwent nodulectomysince 2010. They showed satisfactory healing rates and a limited number of serious complications[17]. A PubMed and gray literature search for surgical debulkingornodulectomy for podoconiosis patient returned only one additional reference[11].

The aim of the current report is to describe a series of patients with lymphedema and nodules due to podoconiosis who underwent nodulectomy in a referral hospital in the Amhara region of Ethiopia between October 2015 and September 2019. We document time to reepithelialisation after nodulectomy, change in number of ADLA episodes, change in quality of life, and recurrence of nodules one year after surgery.

# Material and methods

## **Ethics statement**

Permission to conduct the study was obtained from the teaching hospitals. Local ethics approval was received from the Bahir Dar University College of Medicine and Health Science Ethics review board, reference number 0250/2019, Bahir Dar, Ethiopia. Since nodulectomy and the follow-up management are considered routine care and we used retrospective data without patient identifiers, informed patient consent was not deemed necessary. For the publication of patient photos, oral informed consent was obtained, and any form of patient identification was avoided.

# Study design, setting and population

This is a retrospective study reporting on a series of 21 patients with podoconiosis who underwent nodulectomy in one of two teaching hospitals in Bahir Dar, Ethiopia between January 2015 and December 2019.

Ethiopia is a located in East Africa and has a population of about 103 million. The country is a federal state with nine regional and two special administrations [18], and is challenged with limited health service coverage (39% of the population) [14]. The Amhara regional state lies in the north west of the country and has approximately 21 million inhabitants, 85% of whom live in rural settings [18]. This region has a podoconiosis prevalence of 3.7% and shoe-wearing is strongly advised and promoted as primary and secondary prevention [19,20]. There are two teaching hospitals in Bahir Dar, the capital city of the region: The Felege Hiwot Specialized Hospital and Tibebe Ghion Specialized Hospital. These two hospitals each have about

1500outpatient consultations daily. They both have a specialized dermatology clinic attending cases referred from district hospitals in the region and neighbouring areas. Among other services, the dermatology clinics provide surgical nodulectomy for patients with podoconiosis. Patient with podoconiosis and nodules who have been given nodulectomy service ware recruited from the general dermatology clinics.

# **Diagnosis of podoconiosis**

In the referral hospitals, we used locally developed clinical diagnostic criteria for podoconiosis combining major and minor criteria (see Box 1). All participants were examined by consultant dermatologists using these criteria.

# Nodulectomy procedure and patient follow-up

Nodulectomy is a surgical reduction of redundant skin/swellings on the foots and lower leg. Nodulectomy was performed under local anaesthesia in a surgical theatre. Fibrotic nodules and tumours were excised with the narrowest possible base. Redundant skin was removed and haemostasis achieved using simple gauze compression and artery forceps. The excision surface area was measured and the weight of the removed tissue was noted. Subsequently, the surgical field was cleaned with normal saline and dressed with sterile gauze and compression bandaging to facilitate healing. The wound was left to heal by secondary intention. Antibiotics and analgesics were given at the surgeon's discretion. In patients with extensive nodules, nodulectomy was temporised in multiple subsequent procedures. In addition, patients were instructed on foot hygiene, skin care, bandaging (for those with water bag lymphedema), exercises, and use of socks and shoes. Patient follow-up took place two, four and eight weeks after nodulectomy, and subsequently every three months for one year. Digital photos were taken before and immediately after nodulectomy and on each follow-up visit.

#### Endpoints

We documented the following endpoints: time until re-epitelization, presence of complications (e.g. occurrence of wound infection and non-healing ulcer), number of episodes of ADLA over a period of three months, Dermatology Quality of Life Index (DQLI) at two months and at six months after surgery compared with DQLI at enrolment (before surgery), and recurrence of nodules in the course of the first year after surgery.

An ADLA episode was defined as two or more of the following symptoms or signs: redness, pain, or swelling of the leg or foot, with or without the presence of fever or chills [21]. The number of ADLA episodes was recorded on each follow-up visit.

The DLQI is a widely used dermatology-specific questionnaire consisting of 10 questions concerning health-related quality of life for people with a skin condition[22,23]. In this Ethiopian setting, we used an Amharic version (local language) of the questionnaire [23]. We performed a DLQI assessment prior to the first nodulectomy and 2and 6 months after the last surgical nodulectomy. The DLQI score can be interpreted as follows: 0–1, 'no effect at all'; 2–5, 'small effect'; 6–10, 'moderate effect'; 11–20, 'very large effect'; 21–30, 'extremely large effect' on quality of life [24–26]. Based on an overview on the use of DLQI in various dermatological diseases, we considered that a decrease in DLQI score of 5 points after surgery indicated a meaningful improvement in quality of life[27].

# Box 1. Clinical diagnostic criteria for podoconiosis

#### **Clinical definition**

Podoconiosis is a non-communicable geochemically induced tropical form of lymphedema resulting in bilateral swelling of the lower legs (below-knee) and impaired skin barrier function, acquired through prolonged barefoot exposure to red clay soils of volcanic origin [8].

#### **Diagnostic criteria**

#### Major criteria

- 1. Below-knee lymphedema
- 2. Residence in endemic area during development of lymphedema
- 3. Barefooted walking for prolonged periods of time(more than 10 years)
- 4. Mossy foot in slipper pattern

#### Minor criteria

- 1. Nodules
- 2. Toe fusion
- 3. Both feet affected
- 4. Positive Stemmer's sign test (i.e., the thickened fold of skin at base of the second toe that can be pinched and lifted. This test is positive when the skin cannot be lifted, showing presence of lymphedema)
- 5. Burning sensation on the feet
- 6. Family history of the same illness
- 7. Problem started in the first three decades of life

#### Diagnosis

- 1. *Definitive podoconiosis*: 3 major or 2 major plus 2 minor or 1 major plus 5 minor criteria
- 2. Probable podoconiosis: 2 major or 1 major plus 2 minor or 5 minor criteria

#### Data collection and analysis

Patient data were extracted directly from the patient file and health information management forms, and entered into an Excel data base. The following information was extracted: demographic characteristics, start date and duration of edemaand nodules, staging, history of ADLA episodes, the surgical procedure, antibiotic use and wound care during follow-up, signs of infection, duration of re-epithelization, and recurrence of nodules. To prevent data entry errors, the data was subsequently cross-checked the patient files and Excel database systematically. We computed descriptive statistics (proportion, median, range, and interquartile range) to summarise patient characteristics, organised per patient and per surgery episode. We compared DLQI score before surgery, two months and six month after surgery and number of ADLA episodes before surgery and three months after surgery (paired comparison) and used the exact Wilcoxon signed rank test to assess statistical significance (one-sided test with a significance level of 0.05). The null hypothesis was that the values after surgery were equal or worse than before surgery. R software was used for data analysis and visualisation.

## Results

#### Demographic and clinical characteristics of the patient population

Between October 2015andSeptember 2019, 32 new patients underwent nodulectomy in the two referral hospitals. From these, 11 could not be included in the analysis as some essential information was missing. Baseline characteristics of these excluded patients did not show difference from those with we have complete information. Of the21 patients included in the analysis, 16 (76%)were male; and their median age was 30 years(interquartile range (IQR) 22–43 years). The median duration of lymphedema was 15 years (IQR 12–20 years) and the median duration of the presence of nodules was 9years (IQR 7–12). One quarter of the patients (n = 6; 28%) had nodules on both feet. Two thirds of the patients (n = 14; 67%) had stage III lymphedema in the second leg. The median occurrence of ADLA in the last three months before the surgery was three episodes (IQR 3 to 6) (Table 1).

## Characteristics of surgery and postoperative care

The 21 patients underwent 37 operations: ten patients (48%) had one, six (29%) had two, and five (24%) had three operations. For those who had more than one operation, the median time between two consecutive operations was 29 days (IQR 24–33). The median time of re-epithelization was 23 days (IQR 20–27) (Fig 1). Post-operative wound infection occurred after four out of the 37 procedures. Three months after nodulectomy, two patients reported a single episode of ADLA (each). From the 12 patient who has presented for their follow-up one year

Characteristic	Summary		
Sex	Men:16 (76%)		
	Women: 5 (24%)		
Age (years)	Median 30; range 19–76; interquartile range 22–43		
Duration of lymphedema (years)	Median 15; range 5–27; interquartile range 12–20		
Duration of nodules (years)	Median 9; range 4–20; interquartile range 7–12		
Stage	Stage 4 in one foot and stage 3 in the other: 1 (5%)		
	Stage 3 in both feet: 5 (24%)		
	Stage 3 in one foot and stage 2 in the other: 14 (67%)		
	Stage 3 in one foot and stage 1 in the other: 1 (5%)		
Number of episodes of acute dermato-lymphangio-adenitis in three months before surgery	Median 3; range 0–6; interquartile range 3–6		

Table 1. Demographic and clinical characteristics of podoconiosis patient (n = 21) who have undergone surgical nodulectomy in Bahir Dar, Ethiopia between January 2015 and December 2019.

https://doi.org/10.1371/journal.pntd.0009053.t001



Fig 1. A 35-year old gentleman with a 15-year history of bilaterallymphedema, and 20 by 17 cm multiple joining nodules on the dorsum of the right foot with lymphorhea, pre-nodulectomy(Fig 1A and 1B); post-nodulectomy (Fig 1C and 1D).

https://doi.org/10.1371/journal.pntd.0009053.g001

post-surgery, one patient had mild recurrence of the nodules. In all patients, the podoconiosis stage decreased from stage III to stage II, three months following the last surgery (Table 2).

# Change in total score on Dermatology Life Quality Index over time

The median baseline DLQI before nodulectomy was 16 with an IQR of 13 to 19, (classified as very large effect on patient's quality of life). Two months after surgery, the median score was 11 (IQR 9–14) and six months after surgery, the median score dropped to 5 (IQR 4–7). The total DLQI score was significantly better six months after surgery than before surgery (P<0.0001); the improvement was visible in all six domains of the questionnaire (Table 3 and Fig 2). Also the number of ADLA episodes per three months was significantly lower six months after surgery than before surgery than before surgery (P<0.0001).

The overall outcome of the procedure was a foot more able to be fitted into a shoe and more practical to wash and care for (Fig 3).

# Wound infection and antibiotic use

In 21(57%) of the surgeries either topical or systemic antibiotic was given as a prophylaxis. And 6 (16%) patient was given oral medication for pain management after the surgery. Only

Characteristic	Summary   Median 100; range 10–450; interquartile range 50–150		
Weight of resected nodule (g)			
Wound surface after surgery (cm <sup>2</sup> )	Median 62; range 6–180; interquartile range 38–112		
Place of postoperative wound care	Health center: 26 (70%) Hospital: 7 (19%) Self-care at home: 4 (11%)		
Duration postoperative wound healing (days)	Median 21; range 17–42; interquartile range 20–27		
Postoperative pain medication	None: 31 (84%)		
	Tramadol: 4 (11%)		
	Diclofenac: 2 (5%)		
Postoperative wound infection	Yes: 4 (11%)		
	No: 33 (89%)		

Table 2. Characteristics of surgery and postoperative care after nodulectomy (n = 37) for podoconiosis patientin Bahir Dar, Ethiopia between January 2015 and December 2019.

https://doi.org/10.1371/journal.pntd.0009053.t002

		Before surgery median (IQR)	Six months after surgery median (IQR)	Difference before-after surgery median (IQR)
Section	Maximum score on that section			
Symptoms and feelings	6	4 (3-5)	1 (1–2)	3 (2-4)
Daily activities	6	2 (2-4)	1 (1-2)	2 (1-2)
Leisure	6	3 (2–5)	1 (1-2)	2 (2-3)
Work and school	3	2 (1-2)	0 (0–0)	2 (1-2)
Personal relationships	6	4 (2–5)	1 (1–2)	2 (1-3)
Treatment	3	1 (1-1)	0 (0–0)	1 (1-1)
Total DLQI score	30	16 (13–19)	5 (4-7)	10 (8–12)

Table 3. Scores on sections of the Dermatology Life Quality Index before and six months after surgery for podoconiosis patientin Bahir Dar, Ethiopia between January 2015 and December 2019.

https://doi.org/10.1371/journal.pntd.0009053.t003

in four (11%) procedures, local wound infections occurred. These infections were diagnosed clinically as there was no access to microbiological investigation. None of the patients developed clinical septicaemia. All wound infections resolved, but there was a clear delay in wound healing: the median time until re-epithelisation time was 21 days in non-infected wounds versus 32.5 days in infected wounds.



Fig 2. Change in total score on DQLI over time for podoconiosis patientin Bahir Dar, Ethiopia between January 2015 and December 2019. Time 0 is before the first surgery. DLQI: Dermatology Life Quality Index.

https://doi.org/10.1371/journal.pntd.0009053.g002



Fig 3. A 20-years old boy with a 5-year history of right leg lymphedema followed by swelling on the foot with ulceration. Pre-nodulectomy (Fig 3A and 3B); post-nodulectomy (Fig 3C and 3D).

https://doi.org/10.1371/journal.pntd.0009053.g003

# Discussion

The findings from this case series reaffirm the safety and effectiveness of surgical nodulectomy in patients with podoconiosis. Authors have written widely on the concept of poor wound healing in lymphedema and various surgical procedures have been proposed to deal significant blood loss, morbidity, infections, permanent disfigurement, and recurrence of symptoms [28,29]. It is assumed that damage due surgery to the lymphatic system impairs the healing of both venous and non-venous wounds[30]. In spite of this assumption, we have demonstrated that surgical nodulectomy can be successful in podoconiosis with acceptable healing rates and an encouraging lack of complications. Most importantly we have seen a significant improvement in patient quality of life following nodulectomy and reduction of ADLA episodes.

Limitations of our study are that we did not exclude filariasis as a cause of edema, lack of an untreated control group and lack of data on the excluded patients. A confounding factor could be that, in addition to the nodulectomy service each patient was advised on daily foot care and use of elastic bandaging which can create a bias in the treatment response. A strength of this study is largest case series reported globally compared to the previous study [14]. Moreover, we also looked the nodulectomy outcomes up to one year after the procedure, including the effect on quality of life and episodes of ADLA.

Rapid wound healing and the absence of major complications is in agreement with our own previous publication[17]. E.W. Price in his 1919's book on podoconiosis also stated that "In cases treated in this way there has been no evidence of re-formation of the nodules" [11]. The large improvement in DLQI score between baseline and six months after surgery indicates that the procedure has a great positive impact on quality of life. The clear reduction of ADLA episodes is also encouraging. In addition to the nodulectomy, lymphedema care and foot care instructions may have attributed to the decrease in ADLA episodes. This finding is also in agreement with the GoLBeT study where simple self-care proofed effective in reducing the frequency and duration of ADLA[13]. Yet, self-care is not a sole option in progressed podoconiosis patient with non-pitting edema and woody-hard fibrous nodules causing an inability to wear foot protection. Also, recurrent disabling bacterial infections are un responsive to self-care [17].

We hope the findings of this study will be used to better understand the role of surgical reduction in the management of lymphedema and give confidence to clinicians to undertake similar life quality enhancing procedures in other settings. We anticipate that these nodulectomy outcomes in patients with lymphedema secondary to podoconiosis may be repeated in other tropical lymphedemas such as secondary to filariasis. This may benefit the patients

directly in the future. The result will also give important information for the Ministry of Health and partners in surgical management of podoconiosis and contribute to improving the National (MOH) podoconiois control guidelines. We recommend future multi-cantered case control studies, with appropriate case definition (e.g. exclusion of filariasis) to validate the current result and address some of the limitations of this study.

In conclusion, this study showed that a simple, resource-appropriate, surgical debulking procedure resulted in significant cosmetic, functional, and self-reported quality of life improvements among people affected with podoconiosis.

# Acknowledgments

We thank all the study participants for their willingness to participate in this study. We also acknowledge staff of the dermatology clinic in FelegeHiwotReferal Hospital and TibebeGhion Specialized Hospital for their assistant in the podoconiosis patient care.

# **Author Contributions**

Conceptualization: Wendemagegn Enbiale.

Data curation: Wendemagegn Enbiale, Melesse Gebeyehu.

Formal analysis: Kristien Verdonck.

Funding acquisition: Henry J. C. de Vries.

Investigation: Wendemagegn Enbiale, Melesse Gebeyehu.

Project administration: Wendemagegn Enbiale.

Resources: Henry J. C. de Vries.

Supervision: Henry J. C. de Vries.

Validation: Kristien Verdonck, Johan van Griensven, Henry J. C. de Vries.

Visualization: Wendemagegn Enbiale, Kristien Verdonck, Johan van Griensven, Henry J. C. de Vries.

Writing - original draft: Wendemagegn Enbiale.

Writing – review & editing: Kristien Verdonck, Melesse Gebeyehu, Johan van Griensven, Henry J. C. de Vries.

#### References

- Price EW. The association of endemic elephantiasis of the lower legs in East Africa with soil derived from volcanic rocks. T Roy Soc Trop Med H. 1976; 70(4):288–95. <u>https://doi.org/10.1016/0035-9203</u> (76)90078-x PMID: 1006757
- 2. Price EW, Bailey D. Environmental factors in the etiology of endemic elephantiasis of the lower legs in tropical Africa. Trop Geogr Med. 1984 Mar; 36(1):1–5. PMID: 6328708
- Price EW. A possible genetic factor in non-filarial elephantiasis of the lower legs. Ethiop Med J 1972; 10:87–93 PMID: 4670473
- Davey G, Gebrehanna E, Adeyemo A, Rotimi C, Newport M, Desta K. Podoconiosis: a tropical model for gene-environment interactions? Trans R Soc Trop Med Hyg 2007; 101:91–96 <a href="https://doi.org/10.1016/j.trstmh.2006.05.002">https://doi.org/10. 1016/j.trstmh.2006.05.002</a> PMID: 16884751
- TekolaAyele F, Adeyemo A, Finan C, Hailu E, Sinnott P, Burlinson ND, et al. HLA class II locus and susceptibility to podoconiosis. N Engl J Med. 2012 Mar 29; 366(13):1200–8. https://doi.org/10.1056/ NEJMoa1108448 PMID: 22455414

- Deribe K, Simpson H, Cano J, Pigott D. M., Weaver N. D., Cromwell E. A., et al. Mapping the global distribution of podoconiosis: Applying an evidence consensus approach. PLoSNegl Trop Dis. 2019; 13 (12):e0007925. https://doi.org/10.1371/journal.pntd.0007925 PMID: 31790408
- Deribe K, Cano J, Giorgi E, Pigott D. M., Golding N., Pullan R. L., et al. Estimating the number of cases of podoconiosis in Ethiopia using geostatistical methods. Wellcome Open Res 2017; 2:78 <u>https://doi.org/10.12688/wellcomeopenres.12483.2 PMID: 29152596</u>
- Tekola F, HaileMariam D, Davey G. Economic costs of endemic nonfilarial elephantiasis in Wolaita Zone, Ethiopia. Trop Med Int Health 2006; 11(7):1136–44. https://doi.org/10.1111/j.1365-3156.2006. 01658.x PMID: 16827714
- Enbiale W, Fuller C, Davey G. Podoconiosis: the forgotten tropical lymphedema. Skinmed. 2012 Jul-Aug; 10(4):206–7. PMID: 23008936
- Bartlett J, Deribe K, Tamiru A, Amberbir T, Medhin G, Malik M, Hanlon C, Davey G. Depression and disability in people with podoconiosis: a comparative cross-sectional study in rural northern Ethiopia. Int Health 2015; 8:124–31. https://doi.org/10.1093/inthealth/ihv037 PMID: 26113669
- 11. Price EW.Podoconiosis:Non-filarial Elephantiasis. Oxford: Oxford Medical.1990.
- Tekola-Ayele F, Embiale WY. Podoconiosis: tropical lymphedema of the lower legs. In: Dermatology and Allergology—Principles and Practice. 1st ed. Hong Kong: iConcept Press Ltd; 2014; 201. <u>https://</u> pdfs.semanticscholar.org/df19/c491c922dba13a894cc8f87db4baec5df89d.pdf.
- Negussie H, Molla M, Ngari M, Berkley JA, Kivaya E, Njuguna P, et al. Lymphoedema management to prevent acute dermatolymphangioadenitis in podoconiosis in northern Ethiopia (GoLBeT): a pragmaticrandomised controlled trial. Lancet Glob Health 2018; 6(7):e795–e803. <u>https://doi.org/10.1016/</u> S2214-109X(18)30124-4 PMID: 29773516
- Tekola F, Ayele Z, Mariam DH, Fuller C, Davey G. Development and testing of a de novo clinical staging system for podoconiosis (endemic non-filarial elephantiasis). Tropical medicine & international health: TM & IH, 2008 Oct; 13(10), 1277–1283.
- Mousley E, Deribe K, Tamiru A, Davey G. The impact of podoconiosis on quality of life in Northern Ethiopia. Health Qual Life Outcomes. 2013; 11:122. Published 2013 Jul 17. <u>https://doi.org/10.1186/1477-</u> 7525-11-122 PMID: 23866905
- Sikorski C, Ashine M, Zeleke Z, Davey G. Effectiveness of a simple lymphoedema treatment regimen in podoconiosis management in southern ethiopia: one year follow-up. PLoS Negl Trop Dis. 2010; 4(11): e902. Published 2010 Nov 30. https://doi.org/10.1371/journal.pntd.0000902 PMID: 21152059
- Yeshanehe WE, Tamiru A, Fuller LC. Surgical nodulectomies can heal in patients with lymphoedema secondary to podoconiosis in resource-poor settings. Br J Dermatol. 2017 Oct; 177(4):e128–e129. https://doi.org/10.1111/bjd.15420 PMID: 28256715
- Ethiopian Central Statistical Agency. Population Projections for Ethiopia 2007–2037 Addis Ababa July 2013. file:///ICPS-%20Population%20Projection%202007-%202037%20produced%20in%202012.pdf. Accessed 2 March, 2020.
- Molla YB, Tomczyk S, Amberbir T, Tamiru A, Davey G. Podoconiosis in East and west gojam zones, northern ethiopia. PLoSNegl Trop Dis. 2012; 6(7): e1744.
- Lymphatic filariasis and podoconiosis morbidity management and disability prevention guidelines. Addis Ababa: Ethiopian Ministry of Health; 2016.
- 21. Mues KE, Deming M, Kleinbaum DG, et al. Impact of a community-based lymphedema management program on episodes of Adenolymphangitis (ADLA) and lymphedema progression—Odisha State, India. PLoSNegl Trop Dis. 2014; 8(9):e3140. Published 2014 Sep 11.
- Lewis V, Finlay AY. Ten years experience of the Dermatology Life Quality Index (DLQI). J Invest DermatolSympProc. 2014; 9: 169–80.
- The Dermatology Life Quality Index, 2007. <u>http://www.dermatology.org.uk/downloads/dlqiquest.pdf</u> Accessed February 2 2020.
- Hongbo Y, Thomas CL, Harrison MA, Salek SM, Finlay AY. Translating the science of quality of life into practice: what do Dermatology Life Quality Index scores mean? J Invest Dermatol. 2005; 125: 659–64. https://doi.org/10.1111/j.0022-202X.2005.23621.x PMID: 16185263
- Henok L, Davey G. Validation of the Dermatology Life Quality Index in patients with podoconiosis in south Ethiopia. Brit J Dermatol. 2008; 159: 903–906.
- Mease PJ, Menter MA. Quality-of-life issues in psoriasis and psoriatic arthritis: outcome measures and therapies from a dermatological perspective. J Am AcadDermatol. 2006; 54: 685–704. <u>https://doi.org/ 10.1016/j.jaad.2005.10.008 PMID: 16546593</u>
- Khilji FA, Gonzalez M, Finlay AY 2002. Clinical meaning of change in Dermatology Life Quality Index scores. Br J Dermatol 147 (Suppl. 62): 50

- 28. Garza R III, Skoracki R, Hock K, Povoski SP. A comprehensive overview on the surgical management of secondary lymphedema of the upper and lower extremities related to prior oncologic therapies. BMC Cancer. 2017; 17(01):468. https://doi.org/10.1186/s12885-017-3444-9 PMID: 28679373
- Farina R. III [Elephantiasis of the legs; treatment by dermofibrolipectomy followed by free skin grafts] An Paul Med Cir. 1953; 65(04):299–306.
- WHO, Wound and lymphoedema management, 2011.: https://www.who.int/lymphatic\_filariasis/ resources/9789241599139/en/. Accessed August 28, 2020.